Author’s response to reviews

Title: Timing of first antenatal care attendance and associated factors among pregnant women in public health institutions of Axum Town, Tigray, Ethiopia, 2017: A mixed design study

Authors:

Berhanu Gebrehiwot (brhangs19@gmail.com)

Tilahun Belete (tilahunbe100@gmail.com)

Weyzer Tilahun (weyzer.tilahun@gmail.com)

Betell Tafere (betiela03@gmail.com)

Senait Gebremeskel (senaitgs42@gmail.com)

Version: 4 Date: 30 Aug 2019

Author’s response to reviews:

Point by point response to reviewers and editors

Title: Timing of first antenatal care attendance and associated factors among pregnant women in public health institutions of Axum Town, Tigray, Ethiopia, 2017: A mixed design study

Dear reviewers,

Sorry for delaying to submit our revised manuscript; it is due to lack of internet access and short time interval. Thank you for your constructive comments and suggestions. We found your comments crucial and we have addressed the comments raised and incorporated in to the document. Moreover, kindly find below the responses given to each of the comments raised.

Reviewer 1

Comment #1: Please provide an Abstract after your Title Page. The Abstract of the manuscript should not exceed 350 words and must be structured into separate sections: Background, the context and purpose of the study; Methods, how the study was performed and statistical tests used; Results, the main findings; Conclusions, brief summary and potential implications. Please minimize the use of abbreviations and do not cite references in the abstract.
Response: Dear reviewer, thank you for the comment given and comment is well taken. We have incorporated the abstract after the title page as your recommendation and we have tried to avoid the things you mentioned in the abstract part.

With regards!

Yours sincerely,

On behalf of the co-authors

Berhanu Gebresilassie, MSc